Author’s response to reviews

Title: Hemiarthroplasties in young patients with osteonecrosis or a tumour of the proximal femur; an observational cohort study

Authors:

Pim W van Egmond (p.w.van_Egmond@lumc.nl)
Huub J.L. van der Heide (h.j.l.van_der_Heide@lumc.nl)
Antonie H.M. Taminiau (a.h.m.Taminiau@lumc.nl)

Version: 2 Date: 14 April 2012

Author’s response to reviews:

Dear editor in chief,

We have made the following revisions in our manuscript:
- restructure of our abstract, now containing a method section
- a statement of ethical approval
- a statement of the absence of acknowledgements

We hope that this manuscript will be successfully considered for publication in BMC Musculoskeletal Disorders, and look forward to your reply.

On behalf of all co-authors, Yours Sincerely,

H.J.L van der Heide, MD PhD,
Consultant orthopaedic surgeon